Author's response to reviews

Title: Risk assessment for Thai population: Benchmark dose of urinary and blood cadmium levels for renal effects by hybrid approach of inhabitants living in polluted and non-polluted areas in Thailand

Authors:

Muneko Nishijo (ni-koei@kanazawa-med.ac.jp)
Yasushi Suwazono (suwa@faculty.chiba-u.jp)
Werawan Ruangyuttikarn (ruangyuttikarn@gmail.com)
Kowit Nambunmee (nunt408@hotmail.com)
Witaya Swaddiwudhipong (swaddi@hotmail.com)
Kazuhiro Nogawa (nogawa@chiba-u.jp)
Hideaki Nakagawa (hnakagaw@kanazawa-med.ac.jp)

Version: 4
Date: 26 May 2014

Author's response to reviews:

Dear Editor,

We revised the manuscript based on the comment of the editor to clarify informed consent statement; 1) All subjects gave informed consent prior to participation. 2) It was written or verbal.

We indicated the changes in the text by highlighting so that they are easily identifiable by Editor.

We very much appreciate your consideration of our manuscript for publication in BMC Public Health.

With my best regards

Sincerely yours,

On behalf of the authors:

Muneko Nishijo